**Author's response to reviews**

**Title:** A cross-sectional study of HIV and syphilis infections among male students who have sex with men (MSM) in northeast China: implications for implementing HIV screening and intervention programs

**Authors:**

Jun-jie Xu (xijbeijing@gmail.com)
Kathleen Heather Reilly (ktreilly.ncaids@gmail.com)
Chun-Ming Lu (chunminglu126.com)
Ning Ma (maning_m2000@163.com)
Min Zhang (minzhang100@hotmail.com)
Zhen-Xing Chu (laochu323@yahoo.com.cn)
Jun-jie Wang (jiw405047@sohu.com)
Ke Yun (yunke032433@gmail.com)
Hong Shang (hongshang100@hotmail.com)

**Version:** 5  **Date:** 30 April 2011

**Author's response to reviews:**

1. We have amended the order of references in the manuscript, currently references 2 and 3 were shown in the references list.

2. We have turn off track changes in the manuscript.

3. Acknowledge and funding information were updated.